Author’s response to reviews

Title: Outcomes of Aortic Aneurysm Surgery in England: A nationwide cohort study using Hospital Admissions Data from 2002 to 2015

Authors:

Ahmed Aber (a.aber@sheffield.ac.uk)
Thai Son Tong (t.s.tong@sheffield.ac.uk)
Jim Chilcott (j.b.chilcott@sheffield.ac.uk)
Ravi Maheswaran (r.maheswaran@sheffield.ac.uk)
Steven Thomas (s.m.thomas@sheffield.ac.uk)
Shah Nawaz (shah.nawaz@sth.nhs.uk)
Jonathan Michaels (j.michaels@sheffield.ac.uk)

Version: 3 Date: 02 Nov 2019

Author’s response to reviews:

Dear James Edward,

Many thanks for your email. All the minor changes suggested by yourself have been made to this manuscript including updating the author contribution, availability of data and funding body sections. I would be grateful if you can kindly help us publish this work in your journal.

Kind Regards,

Ahmed Aber